Author's response to reviews

Title: Coffee consumption and risk of cancers: a meta-analysis of cohort studies

Authors:

Xiaofeng Yu (xiaofen-yu@sina.com)
Zhijun Bao (xinyin8868@sina.com)
Jian Zou (apollozou@hotmail.com)
Jie Dong (apolloapol@hotmail.com)

Version: 4 Date: 10 March 2011

Author's response to reviews:

Major revisions

Reference 1: A functioning web link to the website has been provided.

Tables: We included the tables as additional files.

Figures: The figures have been made smaller.